Title: Protocol for a prospective magnetic resonance imaging study on supraspinal lower urinary tract control in healthy subjects and spinal cord injury patients undergoing intradetrusor onabotulinumtoxinA injections for treating neurogenic detrusor overactivity

Authors:

Lorenz Leitner (lorenz.leitner@paralab.balgrist.ch)
Matthias Walter (mwalter@paralab.balgrist.ch)
Patrick Freund (pfreund@paralab.balgrist.ch)
Ulrich Mehnert (umehnert@paralab.balgrist.ch)
Lars Michels (lars.michels@usz.ch)
Spyros Kollias (Spyros.Kollias@usz.ch)
Thomas M Kessler (tkessler@gmx.ch)

Author’s response to reviews: see over
Dear Editor of BMC Urology,

We would like to submit the revised manuscript for consideration of publication in BMC Urology:

“Protocol for a prospective magnetic resonance imaging study on supraspinal lower urinary tract control in patients undergoing intradetrusor onabotulinumtoxinA injections for treating neurogenic detrusor overactivity due to spinal cord injury”

This study will identify structural and functional abnormalities in brain networks of supraspinal LUT control in SCI patients with neurogenic detrusor overactivity compared to healthy controls as well as the supraspinal effect of intradetrusor onabotulinumtoxinA injections in humans suffering from SCI using structural and functional MRI techniques in correlation with clinical measurements.

According to the request of Mr. Reymon U. Ampalaya we are pleased to send you in addition to the revised manuscript (edited lines: 302-303) a certification of the grand of our major funding body “Wings for Life, as well as the ethics approval including a translational of the relevant parts directly to “BMCSeriesEditorial@biomedcentral.com”, to facilitate the reviewing process. On demand we send furthermore a copy of the CONSORT checklist even if this is a prospective research study.

Project Title: Protocol for a prospective magnetic resonance imaging study on supraspinal lower urinary tract control in patients undergoing intradetrusor onabotulinumtoxinA injections for treating neurogenic detrusor overactivity due to spinal cord injury

Applicants: Lorenz Leitner,1* Matthias Walter,1* Patrick Freund,1 Ulrich Mehnert,1 Lars Michels,2 Spyros Kollias,2 Thomas M. Kessler1#

Institution/Hospital: 1 Neuro-Urology, Spinal Cord Injury Centre & Research, University of Zürich, Balgrist University Hospital, Zürich, Switzerland
2 Institute of Neuro-Radiology, University of Zürich, University Hospital Zürich, Zürich, Switzerland
Competing interests: The authors declare that they have no competing interests.

Trial Status: The trial is in the recruiting phase at the time of manuscript submission.

Related Articles: No results of this study have been published or submitted to any journal.

Corresponding author: Thomas M. Kessler, MD
Neuro-Urology
Spinal Cord Injury Center & Research
University of Zürich
Balgrist University Hospital
Forchstrasse 340
8008 Zürich
Switzerland
Phone: +41 44 386 38 45
Fax: +41 44 386 39 07
Email: tkessler@gmx.ch

We would be greatly honored, if BMC Urology would accept our manuscript for review and consider for publication.

Yours sincerely,

Lorenz Leitner

Dr. med. Lorenz Leitner
Neuro-Urology
Spinal Cord Injury Center & Research
University of Zürich
Balgrist University Hospital
Forchstrasse 340
8008 Zürich
Switzerland
Tel: +41 44 386 56 50
Fax: +41 44 386 37 31
Email: lleitner@paralab.balgrist.ch